Author's response to reviews

Title: Managing depression in primary care: A meta-synthesis of qualitative and quantitative research from the UK to identify barriers and facilitators.

Authors:

Elizabeth A Barley (elizabeth.barley@iop.kcl.ac.uk)
Joanna Murray (joanna.murray@iop.kcl.ac.uk)
Paul Walters (paul.walters@iop.kcl.ac.uk)
Andre Tylee (andre.tylee@iop.kcl.ac.uk)

Version: 4 Date: 26 May 2011

Author's response to reviews:

BMC Series Editorial Production Team

Re: Managing depression in primary care: A meta-synthesis of qualitative and quantitative research from the UK to identify barriers and facilitators

We have now made your required changes to the format of our manuscript as follows:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)- completed.

Structure: Change 'Introduction' to 'Background'. completed

Structure: Change 'Contributions' to 'Authors' Contributions.' completed

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. completed

References: Please make sure references are cited in numerical order. We notice reference 65 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited. Reference 65 is cited Table 1 (additional file 1) and listed at the end of the references as per your author instructions. I trust this is OK?

Authors’ contributions: Please use initials instead of full name. completed

Minor revisions (we can make these changes for you, although it will speed up...
publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:

AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Appendix: Please remove vertical lines. I have tried to do this, but the table no longer appeared to make sense so I have left it as it was. Please feel free to change if you think there is a better way.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures. I have reduced the size of the boxes as much as possible without the orientation of the arrows changing automatically.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. Completed

I trust this is satisfactory. Please let me know if you require anything else, but please be aware that I shall be on annual leave and not contactable until 7th June 2011. (In an emergency please contact the second author – Joanna Murray).

Kind regards

Elizabeth Barley